Author’s response to reviews

Title: Progressive disseminated histoplasmosis with concomitant disseminated nontuberculous mycobacterial infection in a patient with AIDS from a nonendemic region (California).

Authors:

Julian Choi (julian.j.choi@gmail.com)

Kia Nikoomanesh (kianikoomanesh@gmail.com)

Jusleen Uppal (jusleenuppal@gmail.com)

Sharon Wang (Supra624@gmail.com)

Version: 2 Date: 28 Jan 2019

Author’s response to reviews:

I have edited the consent to publish section and indicated that written consent was obtained for publishing from the patient